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Status 
 Continue to provide Au beam for RHIC operation and NSRL. 
 AGS cold snake test is done last Friday and is ready for use. The  

helium consumption seen in past a few years is fixed and cryo 
system behaves well. 

 AGS and Booster pp setup: started last Thursday.  
 We use the no snake machine to set up instrumentations such as 

tunemeter, polarimeter and vertical damper.  
 Ready to turn on cold snake today.  
 Switch between Au and proton in the injector chain is improving. 

We will soon start to use tape sequencer to do the switch (turn 
on/off gas leak of IPM and cold/warm snake; move polarimeter 
home for Au user..) 
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